Author’s response to reviews

Title: Interatrial block in prediction of all-cause mortality after first-ever ischemic stroke

Authors:

M Baturova (Maria.Baturova@med.lu.se)
Arne Lindgren (mbaturova@mail.ru)
Yuri Shubik (yshubik@mail.ru)
Jonas Carlson (jonas.carlson@med.lu.se)
Pyotr Platonov (Pyotr.Platonov@med.lu.se)

Version: 2 Date: 27 Jan 2019

Author’s response to reviews:

Dear editor!

Thank you very much for the evaluation of our manuscript. We have changed all requied fields in the manuscript according to the decision letter.

1. Please rename the 'Material and methods' heading to 'Methods'.

   Changed.

2. In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

   The changes are made.
3. Please represent authors' names using their initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Changed.

4. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The text is added.

5. Your availability of data and materials section is not acceptable in its current form.

The statement was changed to:

“The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.”

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Uploaded in a clean version.